

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Le, Quynh Lien; Ha, Hanh Hong; Nguyen, Thi Minh Phuong; Doan, Thuy Duong

#### **Article**

The association between upward and downward earnings management and equity liquidity: empirical evidence from non-financial firms listed in Vietnam

Cogent Business & Management

#### Provided in Cooperation with:

Taylor & Francis Group

Suggested Citation: Le, Quynh Lien; Ha, Hanh Hong; Nguyen, Thi Minh Phuong; Doan, Thuy Duong (2023): The association between upward and downward earnings management and equity liquidity: empirical evidence from non-financial firms listed in Vietnam, Cogent Business & Management, ISSN 2331-1975, Taylor & Francis, Abingdon, Vol. 10, Iss. 2, pp. 1-23, https://doi.org/10.1080/23311975.2023.2211789

This Version is available at: https://hdl.handle.net/10419/294438

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





### **Cogent Business & Management**



ISSN: (Print) (Online) Journal homepage: www.tandfonline.com/journals/oabm20

# The association between upward and downward earnings management and equity liquidity: empirical evidence from non-financial firms listed in Vietnam

Quynh Lien Le, Hanh Hong Ha, Thi Minh Phuong Nguyen & Thuy Duong Doan

**To cite this article:** Quynh Lien Le, Hanh Hong Ha, Thi Minh Phuong Nguyen & Thuy Duong Doan (2023) The association between upward and downward earnings management and equity liquidity: empirical evidence from non-financial firms listed in Vietnam, Cogent Business & Management, 10:2, 2211789, DOI: 10.1080/23311975.2023.2211789

To link to this article: <a href="https://doi.org/10.1080/23311975.2023.2211789">https://doi.org/10.1080/23311975.2023.2211789</a>

9	© 2023 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group.
	Published online: 19 May 2023.
	Submit your article to this journal 🗗
ılıl	Article views: 974
Q <sup>\</sup>	View related articles 🗗
CrossMark	View Crossmark data 🗷
	Citing articles: 1 View citing articles 🗗







Received: 16 March 2023 Accepted: 04 May 2023

\*Corresponding author: Thi Minh Phuong Nguyen, National Economics University, 207 Giai Phong Street, Hanoi, Vietnam E-mail: phuongntm@neu.edu.vn

Reviewing editor: Collins G. Ntim, Accounting, University of Southampton, UK

Additional information is available at the end of the article

## ACCOUNTING, CORPORATE GOVERNANCE & BUSINESS ETHICS | RESEARCH ARTICLE

The association between upward and downward earnings management and equity liquidity: empirical evidence from non-financial firms listed in Vietnam

Quynh Lien Le<sup>1</sup>, Hanh Hong Ha<sup>1</sup>, Thi Minh Phuong Nguyen<sup>1\*</sup> and Thuy Duong Doan<sup>1</sup>

Abstract: The current study aims at examining the impact of earnings management on equity liquidity in the Vietnam stock market when considering the direction of earnings management. We used two proxies of equity liquidity, namely quoted spread and effective spread, and analyzed the data using ordinary least squares (OLS), fixed and random effect models (FEM, REM), and regression with generalized least squares (GLS) to determine the most suitable model. The findings of the study indicate that when downward earnings management increases, the quoted spread also decreases, thus the bid-ask spread is low, meaning that the liquidity is high. This can be explained by the fact that when firms conduct downward earnings management to reduce taxes, investors may expect to buy stocks from these firms at a better price, leading to an increase in trading demand on the market. Whereas upward earnings management has no impact on quoted spread. Our results also



Quynh Lien Le

#### ABOUT THE AUTHORS

Quynh Lien Le is a lecturer at the School of Accounting and Auditing, National Economics University, Hanoi, Vietnam. Her research and teaching interests are focused on accounting, earnings management, corporate governance, corporate finance, and liquidity.

Hanh Hong Ha is a lecturer at the School of Accounting and Auditing, National Economics University, Hanoi, Vietnam. Her research and teaching interests are focused on accounting, auditing and insurance.

Thi Minh Phuong Nguyen is a lecturer at the School of Accounting and Auditing, National Economics University, Hanoi, Vietnam. Her research and teaching interests are focused on accounting, lean accounting, management accounting management accounting and financial statement analysis. Thuy Duong Doan is a lecturer at the School of Accounting and Auditing, National Economics University, Hanoi, Vietnam. Her research and teaching interests are focused on accounting, corporate finance and corporate governance.

#### PUBLIC INTEREST STATEMENT

This work examines the effect of upward and downward earnings management on equity liquidity. We used data from 499 non-financial firms listed in Vietnam stock market. Quoted spread and effective spread were used as proxies for liquidity. Our results suggest that when downward earnings management increases, the quoted spread decreases, indicating that the bid-ask spread is low, and liquidity is high. This can be attributed to the expectation that investors will be able to purchase stocks from firms that engage in downward earnings management at a better price, resulting in an increase in trading demand on the market. However, upward earnings management has no effect on quoted spread. Furthermore, our findings indicate that earnings management has no impact on the effective spread. Thus, in the absence of other factors, an increase in earnings management in either upward or downward directions does not influence the difference between buying and selling prices.







© 2023 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. The terms on which this article has been published allow the posting of the Accepted Manuscript in a repository by the author(s) or with their consent.



showed that there is no evidence to suggest that earnings management has an impact on the effective spread. This means that, in the absence of other factors, an increase in earnings management in either upward or downward directions does not affect the difference between buying and selling prices.

Subjects: Corporate Finance; Financial Management; Corporate Governance

Keywords: earnings management; equity liquidity; earnings management and equity liquidity; non-financial firms

#### 1. Introduction

Healy and Wahlen (1999) defined earnings management as the practice of managers using their discretion in financial reporting and structuring transactions to manipulate financial reports in a way that can mislead stakeholders about a company's economic performance or influence contractual outcomes that depend on reported accounting practices. This practice can compromise the quality of earnings reports and disclosure in general. Despite the strong link between accounting disclosure quality and market liquidity, existing empirical research has not firmly established a connection between earnings management and market liquidity. Therefore, it is important for regulators and investors to understand the economic consequences of a firm's disclosure practices and earnings management practices.

Earnings management has the potential to manipulate financial statements to meet targets or present a favorable financial image, potentially affecting contractual relationships. The use of the accrual basis in financial statement preparation makes it difficult to avoid this phenomenon. Managers often resort to earnings management to maximize the utility and market value of the company. Schipper (1989) states that when management intervenes in the preparation process of financial statements for external parties to flatten, raise, or lower profits, it is classified as earnings management practices. Scott (2006) defines earnings management as the accounting policies adopted by managers to achieve specific objectives. In many cases, managers engage in earnings management for their benefit. For example, Bartov and Mohanram (2004) found that managers tend to manipulate earnings to increase earnings over the duration of calling or putting the stock option. Similarly, Leuz et al. (2003) showed that companies with higher degrees of earnings management have higher liquidity costs. Earnings management can also result in lower-quality accounting information disclosure (Dechow & Dichev, 2002), decreasing the willingness of uninformed traders to trade stock. The evidence suggests that earnings management has negative effects on equity liquidity, and liquidity traders acknowledge the involvement of adverse selection costs in earnings management.

Earnings management is a widespread practice in several countries, including Vietnam, which can have an impact on financial statement users. N. H. Dang et al. (2018) found a significant positive relationship between financial information and stock prices for listed firms on the Vietnam Stock Exchange. Leuz et al. (2003) studied earnings management across different countries and found that corporate governance affects earnings management, as insiders use it to protect their interests. This study suggests that companies with higher earnings management tend to have higher market liquidity. Dechow and Dichev (2002) demonstrated that accrual quality has a positive relationship with earnings persistence, highlighting the importance of measuring accrual quality. The empirical evidence suggests that earnings management activity affects market liquidity, with higher levels of inflation leading to greater market liquidity.

Liquidity has been the central topic of numerous studies in the finance literature (Nguyen and Dao, 2022). Research has demonstrated that equity liquidity, which is unpredictable and subject to change over time, can have adverse effects on investor well-being during unfavorable periods. The liquidity of shares is based on corporate profits when the share price rises (Taha et al., 2023).



Galariotis and Giouvris (2007) found that liquidity is correlated between various assets, while Chordia et al. (2001) suggested that monetary expansions are linked to enhancements in stock market liquidity, and fluctuations in liquidity are connected between both stock and bond markets. Pástor and Stambaugh (2003) explored the association between overall market liquidity and stock pricing and discovered that expected stock returns are positively related to how sensitive returns are to fluctuations in overall liquidity.

The influence of stock liquidity on earnings management is not only determined by its magnitude but also by its informational value, as liquidity may increase investor supervision. Nonetheless, studies indicate that actual earnings management can lead to inferior financial disclosures, resulting in increased capital costs for the company (Francis et al., 2008; Kim and Qi, 2010; Kim and Sohn, 2013). Consequently, actual earnings management may have an adverse effect on a company's long-term profitability and competitiveness, with significant implications for its future cash flow (Kothari et al., 2016). While there have been studies on equity liquidity in Vietnam, they have mainly focused on understanding the relationship between equity liquidity and capital structure. For example, L. T. Dang et al. (2019) explored this relationship without examining the impact of earnings management on equity liquidity. Therefore, this research aims to investigate the effects of downward and upward earnings management on equity liquidity in nonfinancial firms listed on the Vietnam stock market. The study will contribute to the literature review by examining the impact of earnings management on equity liquidity in the context of Vietnam, which is a frontier economy. The findings and recommendations of the study are likely to be relevant and applicable to firms operating in similar economic contexts. In this paper, we examine the relationship between both directions of earnings management (upward and downward earnings management) and measures of equity liquidity that are related to information asymmetry. More specifically, we address the following questions: (1) How to measure earnings management of non-financial firms listed on Vietnamese stock market? (2) How to measure equity liquidity of non-financial firms listed on Vietnam stock market? (3) What is the impact of earnings management on equity liquidity of non-financial firms listed on Vietnamese stock market?

In this study, we use a comprehensive dataset from 499 companies over the period of 2010–2020. We obtain data from Vietstock. We obtain data from Vietstock. Our two key variables are earnings management and equity liquidity. To measure a equity liquidity, we rely on two proxies commonly used in literature: When measuring liquidity, we use two proxies: quoted spread and effective spread following the model proposed by Arya et al. (2003). Liquidity measures are estimated by averaging quoted and effective spreads across all trades each day.

Some studies in the world have previously looked at earnings management, but only mentioned earnings management through accounting value without considering the absolute value of earnings management (Al-Shattarat & Ntim, 2021; Bharath et al., 2013; Edmans & Manso, 2011; Edmans, 2009; Nguyen et al., 2021, 2021). Measuring earnings management using the absolute value of discretionary accruals can provide insight into the degree of earnings management, enabling a more accurate evaluation of the impact of other factors on earnings management (Reynold and Francis, 2000; Wang, 2006; Paul Hribar & Craig Nichols, 2007; Zalata et al., 2021). Our study uses the absolute value of accruals as a proxy for earnings management according to the model of Kothari et al. (2015).

Futhermore, previous studies related to earnings management and equity liquidity Hao and Li (2022), Chen et al. (2022) without separating into 2 separate groups including upward and downward earnings management. Upward earnings management refers to firms inflating their earnings, while downward earnings management refers to deflating earnings (Le and Nguyen, 2023). Because the magnitude of risk differs between upward and downward earnings management, investors are expected to assign different weights to these forms when valuing their portfolios. Therefore, the difference between the paper and previous studies related to our topic lies in the fact that the authors separate the data set of discretionary accruals representing earnings



management into two parts: Upward and downward earnings management, then regression model with the independent variable being the absolute value of discretionary accruals for each case and the dependent variable being the equity liquidity. Our study contributes to literature in two ways. First, similar to previous study, this research adds to the existing literature by synthetic a theoretical framework, generalize fully and comprehensively on the measurement methods of earnings management and the effects of earnings management on equity liquidity. Second, the findings of the study indicate that an increase in downward earnings management leads to a decrease in the quoted spread, indicating a low bid-ask spread and high liquidity. This may be due to investors expecting to purchase stocks from such firms at a better price, thus leading to increased trading demand in the market. However, upward earnings management does not have an effect on quoted spread. Moreover, the research also found no evidence of the relationship between either upward or downward earnings management and effective spread.

The remainder of this paper is organized as follows. Section 1 indicates the background to explain why Vietnam is the appropriate context to conduct our study. Section 2 provides the theorical literature reviews. Section 3 reviews the prior empirical literature in this area and develops the main hypotheses of the paper. Section 4 describes our data sources and the variable construction procedure. Section 5 presents empirical evidence of the impact of earnings management on equity liquidity and Section 7 sets forth our conclusion.

#### 2. Background

The economic reform initiated in 1986 led to Vietnam becoming one of the fastest-growing economies in Asia. The reform was a response to saving the country from near bankruptcy after the collapse of the Soviet Union and other communist blocs, and it transformed Vietnam's economy from centrally planned to a more market-oriented one. Under the pressure of financial liberalization and globalization, the financial sector has been constantly developing towards international standards, and there have been significant advancements in the financial landscape in recent decades. Since the establishment of the first stock exchange in Ho Chi Minh City in 2000, the stock market has also experienced significant growth in terms of capitalization and liquidity.

Vietnam is a developing country with a nascent capital market and an inadequate legal framework, making its corporate sector vulnerable to information asymmetry and agency costs, which can hinder firms' ability to access capital markets. In this context, auditing services play a crucial role in ensuring the transparency, timeliness, and appropriateness of financial statements. An unqualified opinion from auditors is essential for maintaining a firm's reputation and prestige.

Earnings management has become a prevalent issue in Vietnam, with multinational companies introducing it to the country, and many domestic enterprises actively engaging in it. This practice produces unreliable accounting profit information that does not accurately represent a company's operational effectiveness. A few examples of this can be seen in Vietnam Gas Corporation (GAS), which showed a post-tax profit difference of up to 646 billion dong before and after auditing in 2018, according to Vietnam Economic Times. Another example is Hoang Anh Gia Lai Joint Stock Company (HAG), which went from a profit of 253 billion dong to a loss of 2,025 billion dong after auditing. These instances have raised concerns about the accuracy of financial reporting and the transparency of information disclosure in Vietnam.

Equity liquidity refers to the degree of ease with which an asset can be converted into cash without any significant loss of value. It measures the ability of an asset to be sold quickly and at a fair price. In the context of the stock market, liquidity refers to how easily a stock can be bought or sold, and it is influenced by various factors such as the frequency of trading, stock price fluctuations, and the time taken to execute a transaction. The liquidity of securities is influenced by various factors, such as firm performance, policies, and investor psychology. Capital markets play a crucial role in economic development by mobilizing funds from both domestic and international sources. Generally, a stock is considered to be liquid if it has a high trading volume. Stock



price fluctuations can affect liquidity, with appreciation generally improving liquidity and depreciation reducing it. The time taken to execute a transaction also plays a crucial role in determining liquidity, and the introduction of a computerized stock trading system has significantly reduced the time required to execute trades, leading to increased liquidity of shares listed on the stock exchange.

The importance of emerging equity markets for investment portfolios and international diversification has garnered significant attention in recent years. Han et al. (2014) argue that as investment in emerging markets continues to grow, studies in the emerging market context become increasingly important. Typically characterized by illiquidity and high volatility, emerging equity markets may offer a better risk-return profile for portfolio managers (Dheeriya & Torun, 2013). Additionally, emerging markets are considered to have lower integration with the global economy, making liquidity a crucial stock attribute that investors need to pay close attention to. However, most of the research on liquidity has primarily focused on developed markets, with limited studies on emerging stock market liquidity. Furthermore, small countries like Vietnam are often overlooked in the literature. Our research on the impact of earnings management on equity liquidity effects of the Vietnamese stock market may present a strong case because poor liquidity is suggested as one of the main reasons why foreign and institutional investors do not invest in this market.

#### 3. Theoretical literature review

Accounting theory highlights the strong connection between accounting disclosure quality and market liquidity (Kim and Verrecchia 2001; Lambert et al. 2007; Leuz and Verrecchia, 2000).

#### 3.1. Information asymmetry theory

According to information asymmetry theory (Glosten & Milgrom, 1985), there is an uneven distribution of information among economic agents in the market. This leads to an increase in the degree of adverse selection in the market and a widening of the price spread. To address this issue, Charoenwong et al. (2011) suggest that reducing information asymmetry could be the basis for the decision to disclose manipulated earnings. They conducted a study on the effects of governance quality, transparency, and accurate information on adverse selection in the Singaporean context. Several published studies have revealed that high-quality information reduces information asymmetry between informed and uninformed investors, promotes investor confidence, and increases trading volume. For example, Cohen et al. (2008) argue that an information-rich environment tends to significantly narrow the bid and ask price gap. Heflin et al. (2005) also found that the publication of high-quality accounting information improves equity liquidity by reducing information asymmetry. Therefore, it can be concluded that reducing information asymmetry is crucial in promoting equity liquidity in the market. High-quality information, transparency, and accurate disclosure of financial information can improve investor confidence and reduce adverse selection costs, which in turn increases trading volume and liquidity. This suggests that firms should prioritize the disclosure of accurate and transparent financial information to maintain investor confidence and ensure a liquid market.

#### 3.2. Signaling theory

Jensen and Meckling's (1976) integration of signaling theory in business analysis demonstrates the existence of opportunistic conflicts of interest. Recent accounting scandals, as pointed out by Chung et al. (2009), demonstrate that earnings management causes significant losses to shareholders and provides a warning that managers often pursue their private interests under the guise of excuses. Earnings management undermines the quality of earnings reporting and disclosure, leading to high information asymmetries. Although information quality can affect the cost of equity, many studies have found that improved information disclosure results in a clear reduction in a firm's costs (Charoenwong et al., 2011). During times of corporate financial reporting crises, agency costs are particularly severe for companies with high discretionary accruals and information asymmetry costs. Investors may doubt the validity of available information, leading to decreased investment decisions, arbitrage, and increased transaction costs (Lesmond, 2005).



Therefore, firms should prioritize the accuracy and transparency of their financial information to maintain investor confidence, avoid opportunistic behavior by managers, and reduce agency and information asymmetry costs. By doing so, they can improve the quality of their earnings reporting and disclosure, which in turn can lead to a reduction in transaction costs and an increase in investor confidence and liquidity.

#### 4. Empirical literature review and hypothesis development

Liquidity is a crucial element in any market, as it enables efficient trading of securities without significantly impacting prices over a short time. Equidity liquidity is related to spread which refers to the difference between the bid price (the price at which a buyer is willing to purchase an asset) and the ask price (the price at which a seller is willing to sell an asset). When the spread is larger, it means that the bid and ask prices are farther apart, indicating that there is less agreement between buyers and sellers on the value of the asset. This can lead to lower trading activity and less liquidity in the market, as buyers may be less willing to buy at the higher ask price, and sellers may be less willing to sell at the lower bid price. On the other hand, when the spread is smaller, it means that the bid and ask prices are closer together, indicating that there is more agreement between buyers and sellers on the value of the asset. This can lead to higher trading activity and greater liquidity in the market, as buyers and sellers are more likely to agree on a price and execute trades.

Another perspective suggests that higher liquidity can alleviate managers' short-term focus by intensifying shareholder monitoring. Improved liquidity enables investors to purchase larger blocks of shares at more favorable prices, leading to more direct supervision by shareholders (Maug, 1998). Moreover, increased liquidity can lead to better indirect monitoring by raising the risk of investor exits, as lower transaction costs resulting from higher liquidity increase the willingness of investors to sell when managerial opportunism is detected (Bharath et al., 2013; Edmans & Manso, 2011; Edmans, 2009).

Liquidity was evaluated by quoted spread and effective spread, as suggested by the Arya et al. (2003). According to Christie et al. (1994), the quoted spread is the difference between the inside ask and inside bid prices. In contrast, the effective spread is determined by multiplying twice the absolute value of the difference between the trade price and the average of the inside bid and ask prices.

#### 4.1. Information transparency and quoted spread

Quoted spread refers to the cost that market orders incur when they are executed at the quoted price without any improvement. During times of elevated information asymmetry, the bid-ask spread tends to widen because uninformed traders tend to pull their orders away from the market, reducing their chances of trading with informed traders (Ali et al., 2016).

Etemadi et al. (2010) define liquidity as the ability of a market to facilitate the buying and selling of large volumes of securities efficiently. Achieving liquidity requires transparency of information, as studies have consistently shown that information transparency is significantly and negatively related to information asymmetry (Diamond & Verrecchia, 1991; Kim and Verrecchia, 2001; Lambert et al., 2007; Petersen and Plenborg, 2006).

Unequal distribution of information among economic agents leads to increased adverse selection in the market and wider bid-ask spreads, which in turn, reduces liquidity (Verrecchia, 2001). Thus, disclosure policies that reduce the cost of information search result in lower transaction costs and higher trading volume, leading to reduced bid-ask spreads and increased liquidity.

Investors are interested in greater transparency because it reduces investment costs by reducing information asymmetries, leading to lower quoted spreads (Kim and Verrecchia, 2001; Lambert et al., 2007). Studies have found that earnings management increases quote spreads, leading to a decrease in equity liquidity (Arya et al., 2003). Conversely, Diamond and Verrecchia



(1991) demonstrate theoretically that high-quality disclosure reduces information asymmetry between informed and uninformed investors.

Petersen and Plenborg (2006) show that high-quality information reduces information asymmetry in the market, increases investor confidence, and thereby increases trading volume. They also find that firms with high-quality information have lower bid-ask spreads. Lafond et al. (2007) examine the relationship between earnings management, governance, and liquidity and find that better governance practices influence earnings management. They also find a negative relationship between discretionary accruals and liquidity, as measured by bid-ask spreads and trading volume

Poor earnings quality can exacerbate information asymmetries among investors in financial markets. Cheng et al. (2006) find that companies with greater transparency and stronger shareholder rights are associated with a lower cost of equity and assets. Eaton et al. (2007) empirically link disclosure quality measures based on accounting standards with the cost of equity of internationally cross-listed firms. Research shows that increased disclosure is associated with a lower cost of equity.

In conclusion, information transparency is essential for achieving liquidity in financial markets. Disclosure policies that reduce information asymmetry and the cost of information search can result in lower transaction costs and higher trading volume, leading to increased liquidity. We have the following hypotheses:

- H1: Upward earnings management has a positive effect on the quoted spread.
- **H2:** Downward earnings management has a positive effect on the quoted spread.

#### 4.2. Information transparency and effective spread

According to Ajina et al. (2015), the information asymmetry component is a form of compensation that emerges due to the potential loss risk faced by liquidity providers as a result of asymmetric information. Liquidity providers find it difficult to distinguish informed traders and avoid losses when trading with them. To mitigate the possibility of such losses, the effective spread should incorporate an appropriate information asymmetry component, which remunerates liquidity providers for the potential loss risk. This compensation enables liquidity providers to continue their operations even in the presence of informed trading activities. Many studies related to equity liquidity are rooted in Akerlof's (1970) idea of adverse selection, where the cost of adverse selection is reflected in bid-ask spreads in stock offerings to offset expected losses arising from trading with more informed investors. Benston and Hagerman (1974) and Glosten and Harris (1988) explore this concept.

Wasan (2006) examines the impact of information asymmetries among investors on transaction costs based on the difference between call prices and call options. Researchers have also found a relationship between the cost of information and the cost of equity and expected earnings. Amihud and Mendelson (1986) show that higher bid spreads are related to expected earnings, while M. Brennan and Subrahmanyam (1996) demonstrate that variable transaction costs are related to expected income.

Amihud (2002) notes that a measure of liquidity is reflected in expected earnings, and Pástor and Stambaugh (2003) provide evidence that information costs are considered in determining a firm's cost of equity. Lambert et al. (2007) establish a relationship between the quality of accounting information and a firm's cost of goods.



Ross (1977) provides evidence that managers use accounting options to report personal information and company performance. Earnings management is intended to improve financial statements. Regulatory authorities and financial market stakeholders may not be equally receptive to information, and managers may have better information than stakeholders. Ahmed et al. () suggest that conservative management can be used in an opportunistic approach, and manipulated accounting numbers may serve as a market signaling tool.

Arya et al. (2003) challenge the notion that earnings management reduces information transparency, leading to a decrease in equity liquidity. Overall, these studies shed light on the impact of information asymmetries on the cost of equity and the importance of transparency and accurate disclosure in financial markets. We have the following hypotheses:

H3: Upward earnings management has a positive effect on efficiency spread.

**H4:** Downward earnings management has a positive effect on efficiency spread.

#### 5. Research design

#### 5.1. Research data

Consistent with the prior literature, we focus on non-financial firms listed on the Vietnamese stock market, with 499 such firms included in the final sample after excluding financial firms from our sample because these firms are subject to special regulations on financing policies and companies with unavailable data and exceptions. Data was collected from two sets of daily and annual data: (1) daily data relating to stock prices, trading volume, and bid and ask prices, which consists of 1,333,709 observations by day over 11 years from 2010–2020, collected from the Vietstock database; (2) year-by-year data after filtering out ineligible observations, resulting in 3,859 observations for 11 years from 2010–2020.

#### 5.2. Research method

To evaluate the stock liquidity, we utilized two proxies, namely the quoted spread and the effective spread, as suggested by Arya et al. (2003), Ascioglu et al. (2012) and Bafghi et al. (2014). Earnings management was measured using one proxy, namely the absolute discretionary accrual (Kothari et al., 2015). Control variables such as trading volume, volatility, price, return and firm size were also included to ensure more accurate results (Ajina et al., 2015; Dennis & Weston, 2001; Espinosa et al., 2008).

A quantitative research method was used, with Stata 16 used to determine the regression coefficient. The least squares method was used in the Pooled-OLS model, but with panel data, unique characteristics of each entity that can affect the explanatory variable but are not observable needed to be considered. To account for this, two models were employed: fixed effect models (FEM) and random effect models (REM).

Correlation analysis was also conducted to determine the relationship between the quantitative variables in the model. The results showed that dependent variables, independent variables, and control variables were highly correlated, indicating a relationship between the variables. The analysis also helped identify any violation of the assumption of linear regression analysis, such as multicollinearity or high correlation among the independent variables. Therefore, regression analysis was carried out using generalized least squares (GLS) estimation.

Greene (2005) proposed that in models related to earnings management, there may be endogeneity issues between variables. To address the endogeneity problem, we used the System GMM model by Arellano and Bond (1991). One of the advantages of the System GMM model is that it is easier to select instrumental variables because it uses exogenous variables from different time periods or lagged



variables that can be used as instrumental variables for endogenous variables at the current time. As a resuls, the System GMM model provided many instrumental variables to easily satisfy the condition of an overidentified instrumental variable (Overidentification of Estimators).

#### 5.2.1. Empirical models

Based on the literature and identified research gap, we constructed empirical models to investigate the impact of upward and downward earnings management on equity liquidity, presented in the following form (Table 1):

Table 1. Va	riables in the research m	odel		
No	Variable name	Symbol	Measure	References
I	Dependent variable	(ILLIQUIDITY)		
1	Quoted spread	QTSP	Difference between the ask and bid quotes relative to the mid-quote	Arya et al. (2003), Ascioglu et al. (2012), Bafghi et al. (2014)
2	Effective spread	EFSP	Twice the absolute value of the difference between the transaction price and the midquote divided by the mid-quote	Arya et al. (2003) Ascioglu et al. (2012), Bafghi et al. (2014)
II	Independent variab	les		
3	Earnings management	EM	The absolute value discretionary accrual.	Jones (1991); Dechow et al. (1995), Kothari et al., (2005)
III	Control variable		·	
4	Volatility	VOLAT	The annual average standard deviation of equity return	Eaton et al. (2007), Ajina et al. (2015)
5	Trading volume	VOLM	The annual average daily trading volume	Stoll (1978), Kim and Verrecchia (2001), Ajina et al. (2015)
6	Price	PRICE	The average daily closing price of each year	Ascioglu et al. (2012), Ajina et al. (2015)
7	Return	RETURN	The mean daily returns of stock of each year	Amihud et al. (1997), M. J. Brennan et al. (1998)
8	Firm size	SIZE	The natural logarithm of total assets	Lim et al., (2008)

Source: The authors proposed

The data is categorized into two groups based on the direction of Earnings Management (EM): Positive EM denotes upward EM, while Negative EM denotes downward EM.

#### 5.2.2. Equity liquidity measures

According to financial theory (Demsetz, 1968), a deep market is defined by several key characteristics: constantly quoted bid and ask prices, a low bid-ask spread, and small orders. To evaluate market liquidity, two measures are commonly used: quoted spread and effective spread. Quoted spread is the difference between the ask and bid quotes relative to the mid-quote, as shown in



Equation (1). Effective spread is calculated as twice the absolute value of the difference between the transaction price and the mid-quote, divided by the mid-quote, as shown in Equation (2). To account for the positive correlation between dollar-denominated spreads and price, we calculate relative spreads as a proportion of the price. To obtain annual liquidity measures, we first estimate daily liquidity measures by averaging quoted and effective spreads across all trades each day. Then, we average the daily measures across each firm-year, following the model proposed by Arya et al. (2003):

$$Quoted spread = \frac{Askt - Bid_t}{\frac{Askt + Bidt}{2}}$$
 (1)

Effective spread = 
$$2 \frac{\frac{|Askt+Bidt|}{2} - Pt}{\frac{Askt+Bidt}{2}}$$
 (2)

Where:

Bidt: The bid price

Ask<sub>t</sub>: The ask price

The bid and ask quotes used to estimate the liquidity proxies are based on the best bid and offer prices of the firm's market. The greater the spread, the lower the liquidity. To ensure data accuracy, we apply filters similar to those used by Huang and Stoll (1996). The dataset used in this study includes 1,333,709 observations spanning 11 years (2010–2020) and contains information on the closing price adjusted, highest matching price, lowest order matching price, the volume of shares traded, and volume of shares outstanding sourced from the Vietstock database. To obtain annual liquidity measures, we calculate the mean value of both quoted spread and effective spread for each year from 2010 to 2020.

#### 5.2.3. Earnings management measures

Many studies have shown that managers use earnings management mainly to find ways to affect the difference between actual cash flow at the business and profits, creating unexpected accruals usually (DA) on the financial statements. The variable DA represents earnings management, but researchers cannot observe it directly so they estimate it through the determination of the non-discretionary accrual (NDA) (DeAngelo, 1986; Dechow et al., 1995; Healy & Wahlen, 1999).

To detect earnings management, a common approach is to estimate the total accruals (TA) and subtract non-discretionary accruals (NDA) from it.

Total accruals (TA) = Profit after tax—Net cash flow from operating activitie

Total accruals (TA) are divided into 2 parts: Non-discretionary accrual (NDA) and Discretionary accrual (DA). NDA refers to accruals made according to accounting principles, while discretionary accruals (DA) are accruals created by managers to manipulate reported profits. Accruals arise because of a mismatch between the accounting profit presented in the income statement and the net cash flows from operating activities due to the use of two different accounting bases, such as the accrual basis and the cash basis. By identifying and analyzing discretionary accruals, researchers and analysts can determine if a company has engaged in earnings management.

In this research, we used the earnings management model of Kothari *et al.* (2005) to measure the NDA variable because this is the model that proved that when adding the ROA variable based on the Jones (1991) and adjusted Dechow *et al.* (1995) models will increase efficiency and help researchers draw more reliable inferences.



The regression model of the NDA variable according to TA has the following form:

$$\frac{TA_{it}}{A_{it-1}} = \beta_0' \frac{1}{A_{it-1}} + \beta_1' \frac{\Delta REV_{it} - \Delta REC_{it}}{A_{it-1}} + \beta_2' \frac{PPE_{it}}{A_{it-1}} + \beta_3' ROA_i + \varepsilon_{it}$$
(3)

Where:

TA<sub>it</sub>: Total accruals in year t of the company i

DAit: Discretionary accruals in year t of the company i

NDA<sub>it</sub>: Non-discretionary Discretionary accruals in year t of the company i

 $A_{it-1}$ : Total assets of the company i in year t -1

DtREV<sub>it</sub>: Changes in the company's revenue in year t compared to year t-1

DtREC<sub>it</sub>: Changes in the company's receivables in year t compared to year t-1

PPEit: Tangible fixed asset costs of the company i in year t

ROA<sub>it</sub>: Total net profit/Total assets of the company i in year t

 $\beta_0$ ,  $\beta_1$ ,  $\beta_2$ ,  $\beta_3$ : The coefficients indicate the degree of impact of independent variables on dependent variables

 $\varepsilon_{it}$ : Error in the model

Using data including 3,859 observations of 499 firms from 2010 to 2020, the study used Stata 16 software to perform regression according to the Pooled OLS, FEM, REM, and GLS model to find the most suitable model. The equation (3) after regression produced  $\beta_0$ ',  $\beta_1$ ',  $\beta_2$ ', and  $\beta_3$ ' being the estimates of the regression coefficients  $\beta_0$ ,  $\beta_1$ ,  $\beta_2$ ,  $\beta_3$  in the most suitable model. The author replaced  $\beta_0$ ',  $\beta_1$ ',  $\beta_2$ ', and  $\beta_3$ ' in equation (4) to estimate NDAit.

$$NDA_{it} = \beta_0' \frac{1}{A_{it-1}} + \beta_1' \frac{\Delta REV_{it} - \Delta REC_{it}}{A_{it-1}} + \beta_2' \frac{PPE_{it}}{A_{it-1}} + \beta_3' ROA_i + \varepsilon_{it}$$
(4)

• Estimating of discretionary accruals (DA)

After estimating NDA<sub>it</sub>, we used Equation 5) to estimate discretionary accruals DAit:

$$DA_{it} = TA_{it}/TA_{it-1} - NDA_{it}$$
(5)

In the research model, we used the variable EM as a representative of earnings management estimated by the absolute value of the variable DA according to Equation 6:

$$EM_{it} = Abs(DA_{it}) (6)$$

#### 5.2.4. Control variable measures

Volatility is measured by the annual average standard deviation of equity return (Ajina et al., 2015; Eaton et al., 2007; Gregoriou, 2002). The variance of returns is a measure of the unfavorable risk of price fluctuations for investors. Previous studies, such as Heflin and Shaw (2000) and Chae (2005), have reported a negative association between volatility and liquidity. Trading volume: The volume of transactions is measured by the annual average daily trading volume. The impact on liquidity is



ambiguous, with Atiase and Bamber (1994) considering transaction volume a proxy of information asymmetry and Chen et al. (2007) finding a positive relationship between transaction volume and liquidity.

Price: According to the literature, there is a significant relationship between stock price and liquidity. For instance, Attig et al. (2006), Brockman and Chung (2001), and Ajina et al. (2015) have demonstrated a positive correlation between stock price and liquidity.Return: The majority of these models suggest a negative correlation between expected returns and liquidity (Amihud, 2002). In this context, liquidity refers to the liquidity of the equity market. M. J. Brennan et al. (1998) and Fiori (2000) argue that equities with higher liquidity tend to exhibit higher returns.

Firm size is a commonly used control variable in research related to earnings management and liquidity, and is typically measured by the natural logarithm of total assets (Lim et al., 2008).

#### 6. Empirical results and discussion

#### 6.1. Experimental results of the regression model to earnings management

Before conducting regression, we conducted correlation analysis to detect multicollinearity. The results of the correlation analysis are detailed in Table 2:

Table 2. Analys	sis of the correl	ation matrix bet	ween the varial	oles in the mode	el state 1
	TA <sub>t</sub> /A <sub>t-1</sub>	1/A <sub>t-1</sub>	(DtREV <sub>t</sub> - DtREC <sub>t</sub> )/A <sub>t-1</sub>	PPE <sub>t</sub> /A <sub>t-1</sub>	ROA <sub>t-1</sub>
TA <sub>t</sub> /A <sub>t-1</sub>	1				
1/A <sub>t-1</sub>	0.0351	1			
(DtREV <sub>t</sub> -DtREC <sub>t</sub> )/ A <sub>t-1</sub>	0.1435	-0.0436	1		
PPE <sub>t</sub> /A <sub>t-1</sub>	-0.1392	-0.0286	0.073	1	
ROA <sub>t-1</sub>	0.1524	0.0085	0.0338	0.0824	1

The results of Table 2 show that the correlation coefficients between the independent variables in the regression model are all less than 0.5, which means that the model does not have multicollinearity.

Based on the results of the Pooled OLS model regression in Table 3, I found that 3 independent variables are statistically significant. Through the results of empirical analysis after running the Pooled OLS regression, I found that the value F (4, 3854) = 72.57 (sig = 0.000000) helps us to reject the hypothesis Ho that the model has no predictive value and accept the alternative hypothesis H1 that this model exists and can predict the dependent variable ( $TA_t/A_{t-1}$ ) through the independent variables. The adjusted value of  $R^2$  (R-squared = 12.02 %) indicates that 12.02 % of the variation in Total accruals over total assets in year t-1 ( $TA_t/A_{t-1}$ ) can be explained by the independent variables ( $1/A_{t-1}$ , (DtREV-DtREC)/ $A_{t-1}$ , PPEt/ $A_{t-1}$ , ROA $_{t-1}$ ).

From there, I can estimate the discretionary accruals (DA) and finally use the absolute value of DA to estimate the variable EM that earnings management:

$$DA = TA_{t}/A_{t-1} - NDA$$

$$EM = Abs(DA)$$



Table 3. Poo	led OLS mode	l regression r	esults state 1			
TA <sub>t</sub> /A <sub>t-1</sub>	Coef.	Std. Err.	t	P>t	[95% Conf.	Interval]
1/A <sub>t-1</sub>	3.270861	0.441848	7.4	0.000	1	0.987585
(DtREV <sub>t</sub> - DtREC <sub>t</sub> )/A <sub>t-1</sub>	0.052606	0.005483	9.59	0.000	1.01	0.992143
PPE <sub>t</sub> /A <sub>t-1</sub>	-0.03249	0.010083	-3.22	0.001	1.01	0.992291
ROA <sub>t-1</sub>	0.487706	0.030361	16.06	0.000	1.01	0.997301
				Mean VIF	1.01	

Regression results show that the impact of all independent variables on the dependent variable is statistically significant. Finally, I get the model to estimate the non-discretionary accruals (NDA) as follows:

NDA =  $3.270861*1/A_{t-1} + 0.052606*(DtREV-DtREC)/A_{t-1} - 0.03249*PPEt/A_{t-1} + 0.487706*ROA_{t-1}$ 

#### 6.2. Empirical results analyzing the impact of earnings management on the equity liquidity

#### 6.2.1. Descriptive statistics of the variables in the model

To clarify the impact of earnings management on equity liquidity, we conducted a study of 499 non-financial firms over the 11 years from 2010 to 2020 using two datasets at both the daily and yearly levels. We conducted regression analyses using panel data with three methods: Pooled OLS, Fixed Effects Model (FEM), and Random Effects Model (REM).

First, we used statistical analysis to describe the characteristics of the variables in the model. The empirical results in Table 4 provide information on the minimum, maximum, mean, and standard deviation of the dependent variable EM, the independent variables, and the control variables.

The absolute value of the discretionary accrual (EM) has a minimum value of 0.0000203 and a maximum value of 1.633872, the mean value is 0.092276, and the standard deviation is 0.104471. Firms have two tendencies to manage earnings upward and also to manage their earnings downward to achieve their goals. In this research, I use the absolute value of the

Table 4. Descri	iptive statistics	of variables in t	he model		
Variable	Observations	Mean	Standard deviation	Min	Max
Upward Earnings management	1783	0.0969856	0.1235923	0.0000203	1.633872
Downward Earnings management	1965	-0.0880027	0.0832148	-0.9448352	-0.0000452
Earnings management	3748	0.092276	0.104471	0.0000203	1.633872
Quoted Spread	3850	0.025559	0.013523	0	0.102223
Effective Spread	3850	0.377252	0.490351	0	1.976455
Return	3850	0.00053	0.002072	-0.04054	0.011047
Volatility	3850	0.029318	0.009319	0	0.080342
Price	3850	12950.89	15099.52	119.7943	183193.4
Trading Volume	3850	225196.1	828148.5	0.003968	16400000
Firm size	3859	27.46325	1.482934	23.46185	33.67722



discretionary accrual and do not take into the direction of earnings management to accurately reflect the level of earnings management.

**Quoted Spread (QTSP)** is the first metric to measure equity liquidity. The results of the data analysis show that the minimum value of the QTSP variable is 0 and the highest value is 0.10223, the mean value is 0.025559, and the standard deviation is 0.013523.

**Effective Spread (EFSP)** is the second metric to measure equity liquidity. Through data analysis, I find that the minimum value of the EFSP variable is 0 and the highest value is 0.102223, the mean value is 0.377252, and the standard deviation is 0.490351.

**Return (RETURN)** is a measure of the average equity return. Table 4.3 shows that the minimum value of the RETURN variable is -0.04054 and the highest value is 0.011047, the mean value is 0.00053, the standard deviation is 0.002072.

**Volatility (VOLAT)** was determined by the annual average of the standard deviation of equity return. Statistical results show that the minimum value of the VOLAT variable is 0 and the highest value is 0.080342, the mean value is 0.029318, and the standard deviation is 0.009319.

**Price (PRICE)** is the average of the daily closing price of each year. Table 4.3 shows that the minimum value of the PRICE variable is 119.7943 and the highest value is 183,193.4; the mean value is 12,950.89; the standard deviation is 15,099.52. The volatility of the closing price of the adjustment will likely affect the liquidity of the equity.

**Trading Volume (VOLM)** is determined by the annual average of the daily trading volume. Through data analysis, the minimum value of the VOLM variable is 0 and the highest value is 0.102223, the mean value is 0.029318, and the standard deviation is 0.013523.

**Firm Size (SIZE)** is one of the most important indicators that affect equity liquidity. The data shows that the minimum value of the SIZE variable is 23.46185 and the highest value is 33.67722, the mean value is 27.46325, the standard deviation is 1.482934. The difference in firm size is also a factor affecting equity liquidity.

#### 6.2.2. Analyze the relationship between the variables in the model

Before conducting regression, we conducted correlation analysis to detect multicollinearity. Table 5 shows that there exists a linear correlation relationship between the independent variable which is earnings management (EM) and the dependent variable measured by 2 different indicators (QTSP and EFSP), the control variables (RETURN, VOLAT, PRICE, VOLM, and SIZE). The results of Table 5 also show that the correlation coefficients between the independent variables and the control variables in the regression model are mostly less than 0.5, meaning that there is less possibility of multicollinearity.

#### 6.2.3. Regression results according to least squares Pooled OLS model

Based on the results of Pooled OLS model regression in Tables 6, we found that earnings management only affects one dependent variable which is quoted spread. Besides this, in Table 7, we did not find evidence that earnings management affects the remaining dependent variable, namely Effective spread. The value F(6, 3733) = 2769.76 with sig = 0.0000 and F(6, 3733) = 401.22 with sig = 0.0000 means that we reject the hypothesis Ho or the model has no value to predict and accept the alternative hypothesis H1 that this model exists and can predict the dependent variable (QTSP, EFSP) through the independent variables and the control variables. R2-squared values (R-squared = 0.8166 and 0.3921 indicate that 81.66% and 39.21% of the variability in dependent variable (QTSP, EFSP) can be explained by the variations of independent variables (EM) and control variables (RETURN, VOLAT, PRICE, VOLM, SIZE). After conducting OLS regression, we conducted a multicollinearity test. The VIF coefficients of all the independent and control variables are less

Table 5. Correlat	Table 5. Correlation coefficient analysis	alysis						
	Quoted spread	Effective spread	Earnings management	Return	Volatility	Price	Trading Volume	Firm size
Quoted spread	1							
Effective spread	-0.0156	1						
Earnings management	0.0664	-0.0061	<b>~</b>					
Return	0.0543	-0.0583	0.015	1				
Volatility	0.2582	0.0567	6960:0	0.1529	1			
Price	-0.1054	-0.0902	-0.0243	0.0319	-0.194	1		
Trading Volume	0.2096	-0.0858	0.0017	0.0298	-0.0561	-0.0198	1	
Firm size	0.2984	0.0311	-0.0508	-0.0301	-0.3228	0.1894	0.4008	1

Table 6. Pooled OL	Table 6. Pooled OLS regression results for quoted spread	for quoted spread					
	Downw	Downward earnings management	ment	Upwo	Upward earnings management	ment	VIF
Quoted Spread	Coef.	t	P>t	Coef.	1	P>t	
Earnings management	0.000	-0.120	0.905	0.007	2.840	0.005	1.01
Return	-0.116	-0.770	0.440	0.293	2.200	0.028	1.03
Volatility	0.374	12.320	0.000	0.278	9.050	0.000	1.18
Price	0.000	-3.940	0.000	0.000	-3.370	0.001	1.07
Trading Volume	0.000	8.120	0.000	0.000	9.390	0.000	1.21
SIZE	0.001	14.810	0.000	0.001	16.730	0.000	1.37
Number of observations		1962			1777		1.15
R-squared		0.8122			0.8227		



Table 7. Pooled (	OLS regress	ion results	for effecti	ve spread			
		nward earn nanagemen	•	Upward e	arnings ma	nagement	VIF
Effective Spread	Coef.	t	P>t	Coef.	t	P>t	
Earnings management	-0.022	-0.160	0.871	-0.084	-0.910	0.363	1.01
Return	-12.104	-2.080	0.038	-16.430	-3.130	0.002	1.03
Volatility	1.969	1.680	0.094	3.142	2.590	0.010	1.18
Price	0.000	-3.060	0.002	0.000	-4.830	0.000	1.07
TradingVolume	0.000	-4.330	0.000	0.000	-4.320	0.000	1.21
SIZE	0.013	9.190	0.000	0.014	9.470	0.000	1.37
Number of observations		1962			1777		1.15
R-squared		0.3809			0.4056		

than 2, so the model does not have multicollinearity. When performing the white test in two models with quoted spread and effective spread, the results provide the p-values = 0.0000 < 0.05. Thus, both models have heteroskedasticity. Next, we test the autocorrelation through the Wooldridge test. The results have p-values = 0.000 < 0.05. Therefore, the models have autocorrelation. We conducted a regression analysis of the data using both fixed effect models (FEM) and random effect models (REM), while considering both downward and upward earnings management. We used the Hausman test to choose between FEM and REM. However, despite considering the influence of fixed and random factors, our models still exhibited heteroskedasticity and autocorrelation. Therefore, we proceeded to use the general least squares (GLS) method to address these issues (Arellano & Bond, 1991).

6.2.4. Regression results according to generalized least squares regression GLS model Table 8 shows the results of the GLS regression model when considering the impact of earnings management on two dependent variables QTSP and EFSP as follows:

We check for the existence of the endogeneity problem caused by a simultaneous relationship between earnings management and liquidity. To do this, we use the Durbin-Wu-Hausman test of Davidson and MacKinnon (1993). The results of the endogeneity test show prob>0.1 that means earnings management is exogenous (Appendix 1). Therefore, the GLS model is the most suitable model, and the regression results are presented in Table 8.

The results in Table 8 indicate that downward earnings management has a negative effect on the quoted spread. Thus, the bid-ask spread is low, meaning that the liquidity is high. This can be explained by the fact that when firms conduct downward earnings management to reduce taxes, investors may expect to buy stocks from these firms at a better price, leading to an increase in trading demand on the market. Whereas upward earnings management has no impact on the quoted spread. This suggests that when earnings management increases in the downward direction, the quoted spread will decrease if other factors remain constant. This is contradictory to previous research that suggests lower-quality accounting information can lead to concerns among traders about the transparency of accounting information, resulting in reduced equity liquidity in firms with doubts about earnings management. This finding is also different from the research by Ascioglu et al. (2012), which suggests that earnings management impairs the quality of earnings reporting and disclosure in general, leading to high information asymmetry or a disparity in information on the quoted price, and a consequent decrease in equity liquidity.

Table 8. GLS regi	ression results			
	Quote	d spread	Effectiv	e spread
	Downward Earnings management	Upward Earnings management	Downward Earnings management	Upward Earnings management
Earnings management	-0.00183*	-0.001	0.015	0.002
	[-1.72]	[-1.57]	[0.42]	[0.11]
Return	0.287***	0.364***	-3.105**	-2.886*
	[6.53]	[6.04]	[-2.05]	[-1.91]
Volatility	0.683***	0.650***	1.111**	0.776
	[47.96]	[31.13]	[2.35]	[1.56]
Price	0.000***	-0.000***	-0.000***	-0.000***
	[4.41]	[-5.99]	[-4.07]	[-4.91]
Trading Volume	0.000*	0.000***	0.000	0.000
	[1.79]	[2.73]	[0.32]	[0.77]
SIZE	0.000	0.00324***	-0.010	-0.0320***
	[0.99]	[17.98]	[-1.62]	[-4.92]
_cons	0.002	-0.0804***	0.444**	1.072***
	[0.24]	[-15.37]	[2.55]	[5.89]
N	1902	1717	1902	1717

**Note**: t statistics in brackets \* p<0.1, \*\* p<0.05, \*\*\* p<0.01

Table 8 indicates that there is no evidence to support a relationship between downward and upward earnings management and effective spread, indicating that there is no impact of earnings management on equity liquidity through market arbitrage. This means that, in the absence of other factors, an increase in earnings management in either upward or downward directions does not affect the difference between buying and selling prices. The regulation of bid and ask prices of stocks by the market is based on the needs of investors to buy and sell shares. Although earnings management can reduce shareholders' ability to gauge true firm performance, the relationship between earnings management and equity liquidity depends on the nature and components of the total accrual. Non-discretionary accruals reflect the fundamental characteristics of the company and increase transparency, while discretionary accruals reflect managerial caution and can create opacity. Therefore, the fact that managers engage in earnings management may not affect the effective spread between the market bid and ask prices. This finding is in contrast to Dechow and Dichev's (2002) study, which suggested a statistically significant positive relationship between earnings management and effective spread or a wider bid-ask spread created by earnings management to provide price protection, leading to a decrease in equity liquidity. Thus, comparing with the research hypotheses that have been proposed before, the results of testing the research hypotheses are presented in Table 9 as follows:

#### 7. Summary and conclusion

Four proposed hypotheses have been rejected. Downward earnings management has a negative impact on the quoted spread. Therefore, investors need to be cautious when considering investing in firms with earnings management scandals or with doubts about the transparency of financial statements, as these firms may be trying to improve their financial statements or conceal their true financial situation. Moreover, the more firms engage in earnings management, the greater the information asymmetry, and investors should be even more cautious when making transactions because the information they receive may not be accurate.



No	Research hypotheses	Results
H1	Upward earnings management has a positive effect on the quoted spread	Rejected
H2	Downward earnings management has a positive effect on the quoted spread	Rejected
H2	Upward earnings management has a positive effect on the effective spread	Rejected
H4	Downward earnings management has a positive effect on the effective spread	Rejected

Although there is no evidence to confirm that earnings management in both upward and downward directions have an impact on equity liquidity, earnings management reduces investor confidence, leading to a decrease in trading volume. As such, it is recommended that listed firms prioritize long-term benefits by improving the quality of financial reporting information, promoting transparency, accurately reflecting business realities, and adhering to current regulations. Firms should avoid pursuing short-term attractive benefits that may provide false accounting information, as this can lead to a loss of investor confidence. Quoted spread increases, leading to a decrease in the quantity and quality of transactions among investors. As a result, the State Securities Commission should impose appropriate sanctions to deter businesses, individuals, and organizations that intentionally violate regulations or manipulate information on the stock market, which can lead to the loss of confidential information. They should also increase the number of instruments traded on those stock exchanges to promote liquidity (Phan and Vo, 2012). This is essential to maintain a transparent and clean stock market, protecting the interests of investors.

This study is subject to several limitations, including the use of daily data collection to avoid information overload, which may have impacted the accuracy of our analysis. The incomplete data from listed firms also reduced the reliability of our results. Moreover, our measurement methods may not have captured all aspects of each indicator. Due to limitations in time and resources, we were unable to apply multiple models to identify the most appropriate one for the Vietnam stock market. Future research could explore the impact of earnings management on equity liquidity for firms listed on the Vietnamese stock market, utilizing alternative measurement methods, such as real earnings management and M-scores. Furthermore, equity liquidity could be assessed by measuring the ratio between the number of shares being traded and the number of shares outstanding.

#### Funding

This research is funded by National Economics University, Hanoi, Vietnam.

#### Author details

Quynh Lien Le<sup>1</sup> Hanh Hong Ha<sup>1</sup> Thi Minh Phuong Nguyen<sup>1</sup> E-mail: phuongntm@neu.edu.vn

ORCID ID: http://orcid.org/0000-0002-5888-4772

Thuy Duong Doan<sup>1</sup>

<sup>1</sup> National Economics University, Hanoi, Vietnam.

#### Disclosure statement

No potential conflict of interest was reported by the authors.

#### Citation information

Cite this article as: The association between upward and downward earnings management and equity liquidity:

empirical evidence from non-financial firms listed in Vietnam, Quynh Lien Le, Hanh Hong Ha, Thi Minh Phuong Nguyen & Thuy Duong Doan, *Cogent Business & Management* (2023), 10: 2211789.

#### References

Ajina, A., Lakhal, F., & Sougné, D. (2015). Institutional investors, information asymmetry and stock market liquidity in France. International Journal of Managerial Finance, 11(1), 44–59. https://doi.org/10. 1108/IJMF-08-2013-0086

Akerlof, G. A. (1970). The market for "lemons": Quality uncertainty and the market mechanism. *Quarterly Journal of Economics*, 84(3), 488–500. https://doi.org/10.2307/1879431

Ali, S., Liu, B., & Su, J. J. (2016). What determines stock liquidity in Australia? *Applied Economics*, 48(35), 3329–3344. https://doi.org/10.1080/00036846.2015. 1137552

- Al-Shattarat, B., & Ntim, C. G. (2021). The consequence of earnings management through discretionary accruals on the value relevance in Saudi Arabia. Cogent Business & Management, 8(1). https://doi.org/ 10.1080/23311975.2021.1886473
- Amihud, Y. (2002). Illiquidity and stock returns: Cross-section and time-series effects. *Journal of Financial Markets*, 5(1), 31–56. https://doi.org/10. 1016/S1386-41810100024-6
- Amihud, Y., & Mendelson, H. (1986). Asset pricing and the bid-ask spread. *Journal of Financial Economics*, 17(2), 223–249. https://doi.org/10.1016/0304-405X(86) 90065-6
- Amihud, Y., Mendelson, H., & Lauterbach, B. (1997).

  Market microstructure and securities values:

  Evidence from the Tel Aviv stock exchange. *Journal of Financial Economics*, 45(3), 365–390. https://doi.org/10.1016/S0304-405X(97)00021-4
- Arellano, M., & Bond, S. (1991). Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations. *The Review of Economic Studies*, 58(2), 277–297. https://doi.org/10.2307/2297968
- Arya, A., Glover, J., & Sunder, S. (2003). Are unmanaged earnings always better for shareholders? *Accounting Horizons*, 17(s–1), 111–116. https://doi.org/10.2139/ssrn.322260
- Ascioglu, A., Shantaram, P. H., Gopal, V. K., & John, B. M. (2012). Earnings management and market liquidity. Review of Quantitative Finance & Accounting, 38(2), 257–274. https://doi.org/10.1007/s11156-010-0225-9
- Atiase, R. K., & Bamber, L. S. (1994). Trading volume reactions to annual accounting earnings announcements: The incremental role of predisclosure information asymmetry. *Journal of Accounting & Economics*, 17(3), 309–329. https://doi.org/10.1016/ 0165-4101(94)90031-0
- Attig, N., Fong, W. M., Gadhoum, Y., & Lang, L. H. (2006). Effects of large shareholding on information asymmetry and stock liquidity. *Journal of Banking & Finance*, 30(10), 2875–2892. https://doi.org/10.1016/j.jbankfin.2005.12.002
- Bafghi, F. D., Addin, M., & Dehnavi, H. D. (2014). The effect of earnings management on stock liquidity. Interdisciplinary Journal of Contemporary Research in Business, 5(11), 73–82.
- Bartov, E., & Mohanram, P. (2004). Private information, earnings management, and executive stock-option exercises. *The Accounting Review*, 79(4), 889–920. https://doi.org/10.2308/accr.2004.79.4.889
- Benston, G. J., & Hagerman, R. L. (1974). Determinants of bid-asked spreads in the over-the-counter market. *Journal of Financial Economics*, 1(4), 353–364. https://doi.org/10.1016/0304-405X(74)90014-2
- Bharath, S. T., Jayaraman, S., & Nagar, V. (2013). Exit as governance: An empirical analysis. *The Journal of Finance*, 68(6), 2515–2547. https://doi.org/10.1111/jofi.12073
- Brennan, M. J., Chordia, T., & Subrahmanyam, A. (1998).
  Alternative factor specifications, security characteristics, and the cross-section of expected stock returns. *Journal of Financial Economics*, 49(3), 345–373. https://doi.org/10.1016/S0304-405X(98)
- Brennan, M., & Subrahmanyam, A. (1996). Market microstructure and asset pricing: On the compensation for illiquidity in stock returns. *Journal of Financial Economics*, 41(3), 441–464. https://doi.org/10.1016/ 0304-405X(95)00870-K
- Brockman, P., & Chung, D. Y. (2001). Managerial timing and corporate liquidity: Evidence from actual share

- repurchases. Journal of Financial Economics, 61(3), 417–448. https://doi.org/10.1016/S0304-405X(01)
- Chae, J. (2005). Trading volume, information asymmetry, and timing information. The Journal of Finance, 60, 413–442.
- Charoenwong, C., Ding, D. K, & Siraprapasiri, V. (2011). Adverse Selection and Corporate Governance. International Review of Economics & Finance, 20(3), 406–420. https://doi.org/10.1016/j.iref.2010.11.011
- Chen, J. Z., Choy, S. K., Lobo, G. J., & Zheng, Y. (2022).
  Earnings quality and trading volume reactions around earnings announcements: International evidence. *Journal of Accounting, Auditing & Finance, 0* (0), 0148558X2211177. https://doi.org/10.1177/0148558X221117722
- Cheng, C. A., Collins, D., & Huang, H. H. (2006).

  Shareholder rights, financial disclosure and the cost of equity capital. Review of Quantitative Finance & Accounting, 27(2), 175–204. https://doi.org/10.1007/s11156-006-8795-2
- Chen, L., Lesmond, D. A., & Wei, J. (2007). Corporate yield spreads and bond liquidity. The Journal of Finance, 62 (1), 119–149.
- Chordia, T., Roll, R., & Subrahmanyam, A. (2001). Market liquidity and trading activity. *The Journal of Finance*, 56(2), 501–530. https://doi.org/10.1111/0022-1082. 00335
- Christie, W. G., Harris, J. H., & Schultz, P. H. (1994). Why Did NASDAQ market makers stop avoiding odd-eighth quotes? *The Journal of Finance*, 49(5), 1841–1860. https://doi.org/10.2307/2329273
- Chung, H., Sheu, H. J., & Wang, J. L. (2009). Do firms' earnings management practices affect their equity liquidity? *Finance Research Letters*, 6(3), 152–158. https://doi.org/10.1016/j.frl.2009.03.003
- Cohen, D. A., Dey, A., & Lys, T. Z. (2008). Real & accrual-based earnings management in the pre- and post-Sarbanes-Oxley periods. *The Accounting Review*, 83(3), 757–787. https://doi.org/10.2308/accr.2008.83.3.757
- Dang, L. T., Doan, T. N. P., Nguyen, H. T. M., Tran, T. T., Vo, V. X., & McMillan, D. (2019). Analysts and stock liquidity – Global evidence. Cogent Economics & Finance, 7(1), 1625480. https://doi.org/10.1080/ 23322039.2019.1625480
- Dang, N. H., Tran, M. D., & Nguyen, T. L. A. (2018). Investigation of the impact of financial information on stock prices: The case of Vietnam. Academy of Accounting & Financial Studies Journal, 22(2), 1–12.
- Davidson, R., & MacKinnon, J. G. (1993). Estimation and Inference in Econometrics. Oxford University Press. https://doi.org/10.1002/jae.3950100309
- DeAngelo, L. E. (1986). Accounting numbers as market valuation substitutes: A study of management buyouts of public stockholders. *The Accounting Review*, 61(3), 400–420. http://www.jstor.org/stable/247149
- Dechow, P. M., & Dichev, I. D. (2002). The quality of accruals and earnings: The role of accrual estimation errors. *The Accounting Review*, 77(1), 35–59. https://doi.org/10.2308/accr.2002.77.s-1.35
- Dechow, P. M., Sloan, R. G., & Sweeney, A. P. (1995).

  Detecting Earnings Management. The Accounting
  Review, 70(2), 193–225.
- Dechow, P. M., Sloan, R. G., & Sweeney, A. P. (1995). Detecting Earnings Management. *The Accounting Review*, 70(2), 193–225. https://www.jstor.org/stable/248303
- Demsetz, H. (1968). The cost of transacting. Quarterly Journal of Economics, 82(1), 33–53. https://doi.org/ 10.2307/1882244
- Dennis, P. J., & Weston, J. (2001). Who's informed? An analysis of stock ownership and informed trading. An



- Analysis of Stock Ownership and Informed Trading. https://doi.org/10.2139/ssrn.267350
- Dheeriya, P. L., & Torun, E. (2013). Are frontier stock markets more inefficient than emerging stock markets? International Journal of Monetary Economics and Finance, 6(4), 271–284. https://doi. org/10.1504/IJMEF.2013.059943
- Diamond, D., & Verrecchia, R. (1991). Disclosure, liquidity, and the cost of capital. *The Journal of Finance*, 46(4), 1325–1360. https://doi.org/10.1111/j.1540-6261. 1991.tb04620.x
- Eaton, T. V., Nofsinger, J. R., & Weaver, D. G. (2007). Disclosure and the cost of equity in international cross-listing. Rev Quant Finance Acc, 29(1), 1–24. https://doi.org/10.1007/s11156-007-0024-0
- Edmans, A. (2009). Blockholder trading, market efficiency, and managerial Myopia. *The Journal of Finance*, 64 (6), 2481–2513. https://doi.org/10.1111/j.1540-6261. 2009.01508.x
- Edmans, A., & Manso, G. (2011). Governance through trading and intervention: A theory of multiple blockholders. *The Review of Financial Studies*, 24(7), 2395–2428. https://doi.org/10.1093/rfs/hhq145
- Espinosa, J. R., Giudice, G. F., & Riotto, A. (2008).
  Cosmological implications of the Higgs mass measurement. *Journal of Cosmology and Astroparticle Physics*, 2008(05), 002. https://doi.org/10.1088/1475-7516/2008/05/002
- Etemadi, H., Resayian, A., & Kordtabar, H. (2010). The relationship between corporate governance instruments and stocks liquidity. The Journal of Development and Capital, 5, 31–59.
- Fiori, F. (2000). Liquidity premia in the equity markets: An investigation into the characteristics of liquidity and trading activity. Unpublished paper, University of Chicago.
- Francis, J., Nanda, D., & Olsson, P. (2008). Voluntary disclosure, earnings quality, and cost of capital. *Journal of Accounting Research*, 46(1), 53–99. https://doi.org/10.1111/j.1475-679X.2008.00267.x
- Galariotis, E. C., & Giouvris, E. (2007). Liquidity commonality in the London stock exchange. *Journal of Business Finance & Accounting*, 34(1–2), 374–388. https://doi.org/10.1111/j.1468-5957.2006.00664.x
- Glosten, L. R., & Harris, L. E. (1988). Estimating the components of the bid/ask spread. *Journal of Financial Economics*, 21(1), 123–142. https://doi.org/10.1016/0304-405X(88)90034-7
- Glosten, L. R., & Milgrom, P. R. (1985). Bid, ask and transactions prices in a specialist market with heterogeneously informed traders. *Journal of Financial Economics*, 14(1), 71–100. https://doi.org/10.1016/0304-405X(85)90044-3
- Greene, W. (2005). Reconsidering heterogeneity in panel data estimators of the stochastic frontier model. Journal of Econometrics, 126, 269–303.
- Gregoriou, G. N. (2002). Hedge fund survival lifetimes. Journal of Asset Management, 3(3), 237–252. https://doi.org/10.1057/palgrave.jam.2240078
- Han, J., Khapko, M., & Kyle, A. S. (2014). Liquidity with High-Frequency Market Making. Swedish House of Finance Research Paper No. 14-06. https://doi.org/10. 2139/ssrn.2416396
- Hao, G. Q., & Li, K. (2022). Options trading and earnings management: Evidence from the penny pilot program. *Journal of Corporate Finance*, 77, 102290. https://doi.org/10.1016/j.jcorpfin.2022.102290
- Healy, P. M., & Wahlen, J. M. (1999). A review of the earning management literature and its implications for standard setting. Accounting Horizons, 13(4), 365–383. https://doi.org/10.2308/acch.1999.13.4.365

- Heflin, F., & Shaw, K. W. (2000). Blockholder ownership and market liquidity. *Journal of Financial and Quantitative Analysis*, 35(4), 621–633. https://doi.org/ 10.2307/2676258
- Heflin, F., Shaw, K. W., & Wild, J. J. (2005). Disclosure policy and market liquidity: Impact of depth quotes and order sizes. Contemporary Accounting Research, 22(4), 829–866. https://doi.org/10.1506/EETM-FALM-4KDD-9DT9
- Hribar, P., & Craig Nichols, D. (2007). The use of unsigned earnings quality measures in tests of earnings management. *Journal of Accounting Research*, 45(5), 1017–1053. https://doi.org/10.1111/j.1475-679X. 2007.00259.x
- Huang, R. D., & Stoll, H. R. (1996). Dealer versus auction markets: A paired comparison of execution costs on NASDAQ and the NYSE. *Journal of Financial Economics*, 41(3), 313–357. https://doi.org/10.1016/ 0304-405X(95)00867-E
- Jensen, M. C., & Meckling, W. H. (1976). Theory of the Firm: Managerial behavior, agency costs and ownership structure. *Journal of Financial Economics*, 3(4), 305–360. https://doi.org/10.1016/0304-405X(76)90026-X
- Jones, J. (1991). Earnings management during import relief investigations. *Journal of Accounting Research*, 29(2), 193–228. https://doi.org/10.2307/2491047
- Kim, D., & Qi, Y. (2010). Accruals quality, stock returns, and macroeconomic conditions. The Accounting Review, 85(3), 937–978. https://doi.org/10.2308/accr. 2010.85.3.937
- Kim, J. -B., & Sohn, B. C. (2013). Real earnings management and cost of capital. *Journal of Accounting and Public Policy*, 32(6), 518–543. https://doi.org/10.1016/j.jaccpubpol.2013.08.002
- Kim, O., & Verrecchia, R. E. (2001). The relation among disclosure, returns, and trading volume information. The Accounting Review, 76(4), 633–654. https://doi. org/10.2308/accr.2001.76.4.633
- Kothari, S. P., Leone, A. J., & Wasley, C. E. (2005). Performance matched discretionary accrual measures. *Journal of Accounting and Economics*, 39(1), 163–197.
- Kothari, S. P., Leone, A. J., & Wasley, C. E. (2005).
  Performance matched discretionary accrual measures.
  Journal of Accounting & Economics, 39(1), 163–197.
  https://doi.org/10.1016/j.jacceco.2004.11.002
- Kothari, S. P., So, E., & Verdi, R. (2016). Analysts' forecasts and asset pricing: A survey. *Annual Review of Financial Economics*, 8(1), 197–219. https://doi.org/ 10.1146/annurey-financial-121415-032930
- LaFond, R., Lang, M. H., & Skaife, H. A. (2007). Earnings smoothing, governance and liquidity: International evidence. Working paper. The University of North Carolina. https://doi.org/10.2139/ssrn.975232
- Lambert, R., Leuz, C., & Verrecchia, R. (2007). Accounting information, disclosure, and the cost of capital. Journal of Accounting Research, 45(2), 385–420. https://doi.org/10.1111/j.1475-679X.2007.00238.x
- Le, Q. L., & Nguyen, H. A. (2023). The impact of board characteristics and ownership structure on earnings management: Evidence from a frontier market. Cogent Business & Management, 10(1). https://doi. org/10.1080/23311975.2022.2159748
- Lesmond, D. (2005). Liquidity of emerging markets.

  Journal of Financial Economics, 77(2), 411–452.

  https://doi.org/10.1016/j.jfineco.2004.01.005
- Leuz, C., Nanda, D., & Wysocki, P. D. (2003). Earnings management and investor protection: An international comparison. *Journal of Financial Economics*, 69 (3), 505–527. https://doi.org/10.1016/S0304-405X(03) 00121-1



- Leuz, C., & Verrecchia, R. E. (2000). The economic consequences of increased disclosure. *Journal of Accounting Research*, 38(1), 91–124. https://doi.org/10.2307/2672910
- Lim, C., Thong, T., & Ding, K. (2008). firm diversification and earnings management: Evidence from seasoned equity offerings. *Review Quantitative Finance* Accounting, 30(1), 69–92. https://doi.org/10.1007/ s11156-007-0043-x
- Maug, E. (1998). Large shareholders as monitors: Is there a trade-off between liquidity and control? *The Journal of Finance*, 53(1), 65–98. https://doi.org/10. 1111/0022-1082.35053
- Nguyen, N. P. A., & Dao, T. T. B. (2022). Liquidity, corporate governance and firm performance: A meta-analysis. Cogent Business & Management, 9(1). https://doi.org/ 10.1080/23311975.2022.2137960
- Nguyen, H. A., Le, Q. L., & Vu, T. K. A. (2021). Ownership structure and earnings management: Empirical evidence from Vietnam. *Cogent Business & Management, 8*(1). https://doi.org/10.1080/ 23311975.2021.1908006
- Pástor, Ľ., & Stambaugh, R. F. (2003). Liquidity risk and expected stock returns. *Journal of Political Economy*, 111(3), 642–685. https://doi.org/10.1086/374184
- Petersen, C., & Plenborg, T. (2006). Voluntary disclosure and information asymmetry in denmark. *Journal of International Accounting, Auditing & Taxation*, 15(2), 127–149. https://doi.org/10.1016/j.intaccaudtax. 2006.08.004
- Phan, D. N., & Vo, T. H. H. (2012). Determinants of stock market development in southeast Asian Countries. *Journal of Economics and Development*, 44(1), 101–112. https://doi.org/10.33301/2012.14.01.06
- Reynolds, J. K., & Francis, J. R. (2000). Does size matter? The influence of large clients on office-level auditor

- reporting decisions. *Journal of Accounting and Economics*, 30, 375–400.
- Ross, S. A. (1977). The Determination of Financial Structure: The Incentive-Signaling Approach. *Bell Journal of Economics*, 8(1), 23–40. https://doi.org/10.2307/3003485
- Schipper, K. (1989). Commentary on earnings management. Accounting Horizons, 3(4), 91–102.
- Scott, W. R. (2006). Financial Accounting Theory. University of Waterloo.
- Stoll, H. R. (1978). The pricing of security dealer services: An empirical study of NASDAQ stocks. The Journal of Finance, 33(4), 1153–1172. https://doi.org/10.1111/j. 1540-6261.1978.tb02054.x
- Taha, R., Al-Omush, A., & Al-Nimer, M. (2023). Corporate sustainability performance and profitability: The moderating role of liquidity and stock price volatility evidence from Jordan. Cogent Business & Management, 10(1). https://doi.org/10.1080/23311975.2022.2162685
- Verrecchia, R. E. (2001). Essays on Disclosure. Journal of Accounting & Economics, 32(1–3), 97–180. https://doi.org/10.1016/S0165-4101(01)00025-8
- Wang, D. (2006). Founding family ownership and earnings quality. *Journal of Accounting Research*, 44(3), 619–656. https://doi.org/10.1111/j.1475-679X.2006.00213.x
- Wasan, S. (2006). Do accruals exacerbate information asymmetry in the market?. Unpublished doctoral dissertation, Graduate Faculty of the Louisiana State University and Agricultural and Mechanical College.
- Zalata, Zalata, A. M., Ntim, C. G., Alsohagy, M. H., & Malagila, J. (2021). Gender diversity and earnings management: The case of female directors with financial background. Review of Quantitative Finance & Accounting, 58(1), 101–136. https://doi.org/10.1007/s11156-021-00991-4



#### Appendix 1

We conducted the Durbin-Wu-Hausman test by running the following regression:

$$EM = 60 + 61*ROE + 62*LEV + 63*CF + 64*Growth + 65*SIZE + vt$$

Where: ROE is the return on equity, LEV is the leverage (total debt/total assets), CF is the operating cash flow, Growth is the revenue growth rate, and Size is the firm size. Next, we predicted the residual EM\_res =  $v_t$  and ran the following regression:

Liquidity =  $\gamma_0 + \gamma_1 * EM + \gamma_2 * Return + \gamma_3 * Volatility + \gamma_4 * Price + \gamma_5 * Trading Volume + \gamma_6 * SIZE + \gamma_7 * EM_res + u_t$ .

Finally, we tested the hypothesis  $H_0$ :  $\gamma_7$  of EM\_res = 0.

The p-values of the test were 0.8477 when we used Effective-spread for Liquidity and 0.5147 when we used Quoted-spread for Liquidity. As the p-values are greater than 10%, we can conclude that OLS is consistent.